A point-by-point response to the Reviewers’ comments

Reviewer 1.

Thank you very much for the kind words of appreciation for our paper. Concerning the comments:

1. I have divided table 1 into several parts, as suggested

Reviewer 2.

Thank you very much for the kind words of appreciation for our paper. Concerning the comments:

1. I have added new cardiac markers e.g renin, dynorphin, etc, as required